Author's response to reviews

Title: Prolonged survival of a patient with metastatic leptomeningeal melanoma treated with BRAF inhibition-based therapy: a case report

Authors:

Dae Won Kim (dkim4@mdanderson.org)
Edelyn Barcena (ebarcena@broncs.utpa.edu)
Urvi N Mehta (unmehta@mdanderson.org)
Michell L Rohlf (mrohlf@mdanderson.org)
Ashok J Kumar (akumar@mdanderson.org)
Marta Penas-Prado (mpenaspr@mdanderson.org)
Kevin B Kim (kimkb@sutterhealth.org)

Version: 2
Date: 12 July 2014

Author's response to reviews: see over
Dear Editor

Please find enclosed a manuscript entitled: “Prolonged survival of a patient with metastatic leptomeningeal melanoma treated with BRAF inhibition-based therapy: a case report” which we are submitting for exclusive consideration of publication as an article in BMC Cancer. The paper demonstrates clinically meaningful benefit of BRAF inhibitor-based therapy in a patient with a devastating leptomeningeal metastatic disease of melanoma and suggests a need for close investigation of this therapeutic approach in patients with this devastating disease. Therefore, this paper should be of interest to a broad readership including clinicians and translational researchers.

Thank you for your consideration of our work.

Sincerely

Dae Won Kim, MD
Melanoma Medical Oncology Fellow
Department of Melanoma Medical Oncology
The University of Texas MD Anderson Cancer Center, Houston, Texas, USA